Author's response to reviews

Title: Association between H-RAS T81C genetic polymorphism and gastrointestinal cancer risk: a population based case-control study in China

Authors:

Yongjing Zhang (bioinfo406@yahoo.com.cn)
Mingjuan Jin (jinmj@zju.edu.cn)
Bing Liu (liubing7362@sohu.com)
Xinyuan Ma (maxy_zls@yahoo.com.cn)
Kaiyan Yao (yaoky_zls@yahoo.com.cn)
Qilong Li (liql_zls@yahoo.com.cn)
Kun Chen (ck@zju.edu.cn)

Version: 3 Date: 18 June 2008

Author's response to reviews: see over
Dear Iratxe Puebla,

Thank you for your letter regarding our manuscript entitled “Association between H-RAS T81C genetic polymorphism and gastrointestinal cancer risk: a population based case-control study in China” (MS: 1388742149189765). We are very delighted to be informed that the revision of the manuscript was accepted in principle.

This version of manuscript was revised according to Prof. Shen’s suggestion on the meta-analysis. Random effect model was performed to estimate the pooled odds ratio.

In addition, the figure was cropped as closely as possible and the highlighting or tracking were removed form the manuscript text as requested.

We hope that you consider the revised version of manuscript adequate for publication in BMC cancer. We are looking forward to hear from you.

Yours sincerely,

Yongjing Zhang
Department of Epidemiology and Health Statistics,
Zhejiang University School of Medicine